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ABSTRACT 
In the last few decades, greenhouse gas emission has increased very rapidly due to a surge 

in energy demand which results in a rise in global temperature. Hydrogen is an alternative 

source of energy because it is a non-polluting, eco-friendly, and regenerating nature 

which makes it a most attractive fuel source. Thermochemical decomposition of water in 

the Iodine-Sulfur (I-S) cycle for hydrogen production is the most reliable and propitious 

because of its higher efficacy of 56 %. I-S process involves mainly three reactions 

namely, Bunsen reaction, sulfuric acid decomposition, and hydro iodic cycle 

decomposition. Among them, thermal decomposition of sulfuric acid is highly corrosive 

and very energy-intensive usually requires a high temperature of approximately 973-1173 

K and exhibits a large kinetic barrier (73-150 kJ/mol). Finding ways to circumvent the 

high energy consumption may render a path towards enabling a more efficient, and more 

economically viable low-temperature process.  

This research is focused on the estimation of thermodynamic properties including phase 

behavior and chemical speciation, finding a suitable catalyst, and reactor design for high 

energy demanding sulfuric acid decomposition for hydrogen production. In this work, 

metal oxides and spinels are supported over mesoporous silica and pretreated silicon 

carbide. These catalysts were characterized through XRD, FESEM, HRTEM, and BET 

following activity and 100 h long-run stability testing. Iron oxide and spinels dispersed 

on a silica surface grown on a SiC support material are found to be the best catalyst for 

an endothermic high temperature and highly corrosive reaction condition such as sulfuric 

acid decomposition. The combined theoretical and experimental findings not only allow 

us to design cheaper and stable catalysts but also provide an atomistic insight into the 

mechanism of the metal-support interaction. Consequently, the successful tuning of the 

latter helps in the enhancement of overall activity and stability. The high activity and 

stability are ascribed to the presence of oxygen deficit sites in the catalyst developed for 

the model reaction under consideration.  

After that, the reaction kinetics of sulfuric acid decomposition over highly active and 

thermally stable Fe2O3/SiC-Pretrt catalyst is investigated in a fixed-bed quartz reactor. 

To scale-up the process and improve the heat transfer and integration, the influences of 
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the different sizes and shapes of the catalyst pellet on the reaction performance and 

effectiveness factor are estimated. Further, both decomposition and recuperation can be 

achieved simultaneously through effective reactor structure optimization. 

 

 

 

 

 

 

 

 

 

 

 

 

 

 

 

 

 

 

 

 

 

 



                                                               vii 

सारांश 
िपछले कुछ दशको ंम8, ऊजा< की मांग म8 वृAB के कारण Eीनहाउस गैस उKज<न म8 बMत तेजी से 

वृAB Mई है िजसके पPरणामQRप वैिSक तापमान म8 वृAB Mई है। हाइडW ोजन ऊजा< का एक 

वैकAYक Zोत है [ोिंक यह एक गैर-]दूषणकारी, पया<वरण के अनुकूल और पुन: उbc करने 

वाली ]कृित है जो इसे सबसे आकष<क ईंधन Zोत बनाती है। हाइडW ोजन उbादन के िलए आयोडीन-

सfर (आई-एस) चh म8 पानी का थमjकेिमकल अपघटन 56% की उm ]भावकाPरता के कारण 

सबसे िवSसनीय और अनुकूल है। आई-एस ]िhया म8 मुo Rप से तीन ]ितिhयाएं शािमल हp, 

अथा<त् बrन ]ितिhया, सstूPरक एिसड अपघटन, और हाइडW ो आयोिडक चh अपघटन। उनम8 

से, सstूPरक एिसड का थम<ल अपघटन अuिधक संvारक होता है और बMत ऊजा<-गहन 

आमतौर पर लगभग 973-1173 K के उm तापमान की आवxकता होती है और एक बड़ा गितज 

अवरोध (73-150 kJ/mol) ]दिश<त करता है। उm ऊजा< खपत को रोकने के तरीके खोजना एक 

अिधक कुशल, और अिधक आिथ<क Rप से {वहाय< कम तापमान ]िhया को सvम करने की 

िदशा म8 एक माग< ]|ुत कर सकता है। 

यह शोध चरण {वहार और रासायिनक िविश}ता सिहत थमjडायनािमक गुणो ंके आकलन पर 

क8 ि~त है, हाइडW ोजन उbादन के िलए सstूPरक एिसड अपघटन की मांग करने वाली उm ऊजा< 

के िलए एक उपयु� उ�ेरक और Pरए�र िडजाइन की खोज करता है। इस काम म8 मेसोपोरस 

िसिलका और ]ेटW े टेड िसिलकॉन काबा<इड के ऊपर मेटल ऑ�ाइड और A�न� को सहारा िदया 

जाता है। इन उ�ेरको ंको ए�आरडी, एफईएसईएम, एचआरटीईएम, और बीईटी िन�िलAखत 

गितिविध और 100 घंटे लंबे समय तक चलने वाली A�रता परीvण के मा�म से िचि�त िकया गया 

था। एक SiC समथ<न सामEी पर उगाए गए िसिलका सतह पर फैले आयरन ऑ�ाइड और 

A�न� को एंडोथिम<क उm तापमान और अuिधक संvारक ]ितिhया A�ित जैसे सstूPरक 

एिसड अपघटन के िलए सबसे अ�ा उ�ेरक पाया जाता है। संयु� सैBांितक और ]ायोिगक 

िन�ष< न केवल हम8 स|े और A�र उ�ेरक िडजाइन करने की अनुमित देते हp, बA� धातु-

समथ<न बातचीत के तं� म8 एक परमाणु अंत�<ि} भी ]दान करते हp। नतीजतन, बाद की सफल 

�ूिनंग समE गितिविध और A�रता को बढ़ाने म8 मदद करती है। उm गितिविध और A�रता को 

िवचाराधीन मॉडल ]ितिhया के िलए िवकिसत उ�ेरक म8 ऑ�ीजन की कमी वाली साइटो ंकी 

उपA�ित के िलए िज�ेदार ठहराया गया है। 
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उसके बाद, अuिधक सिhय और ऊ�ीय Rप से A�र Fe2O3/SiC-Pretrt उ�ेरक पर 

सstूPरक एिसड के अपघटन की ]ितिhया कैनेटी� की जांच एक िनि�त-बेड  ाट्<ज Pरए�र 

म8 की जाती है। ]िhया को बढ़ाने और गम¡ ह|ांतरण और एकीकरण म8 सुधार करने के िलए, 

]ितिhया ]दश<न और ]भावशीलता कारक पर उ�ेरक गोली के िविभc आकारो ंऔर आकारो ं

के ]भावो ंका अनुमान लगाया जाता है। इसके अलावा, ]भावी Pरए�र संरचना अनुकूलन के 

मा�म से अपघटन और ऊजा< पुनः ]ाA£ दोनो ंको एक साथ ]ा£ िकया जा सकता है। 
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